Author's response to reviews

Title: Perspectives on child diarrhoea management and health service use among ethnic minority caregivers in Vietnam

Authors:

Thilde Rheinländer (thtr@sund.ku.dk)
Helle Samuelsen (H.Samuelsen@anthro.ku.dk)
Anders Dalsgaard (ad@life.ku.dk)
Flemming Konradsen (flko@sund.ku.dk)

Version: 7 Date: 31 August 2011

Author's response to reviews:

Only change made:

I have formatted tabel 2 and 3 (page 32 and 33) to fit into each one page, since I was not satisfied with cells jumping over 2 pages.

Outstanding issue:

The order of the authors must be correct and correspond to the title page of the manuscript; the on-line system has STILL NOT registered the 2nd author. PLEASE CORRECT!

Regards,

Thilde Rheinländer (corresponding author)